Author’s response to reviews

Title: Customized pachymetric guided epithelial debridement for corneal collagen cross linking

Authors:

George D Kymionis (kymionis@med.uoc.gr)
Vasilios F Diakonis (diakonis@gmail.com)
Efekan Coskunseven (efekan.coskunseven@dunyagoz.com)
Mirko Jankov (mirkojankov@mac.com)
Sonia H Yoo (syoo@med.miami.edu)
Ioannis G Pallikaris (pallikar@med.uoc.gr)

Version: 4 Date: 26 July 2009

Author’s response to reviews: see over
Vasilios F. Diakonis, M.D.
University of Crete,
Department of Medicine,
Institute of Vision and Optics (BEMMO),
71110 Voutes, Heraklion, Crete,
Greece. 

Tel: +30 2810 392450
Fax: +30 2810 394653
E-mail: diakonis@gmail.com

Ms No. 2061028657220805 BMC Ophthalmology,
“Customized pachymetric guided epithelial debridement for corneal collagen cross linking”

Dear Rikki Graham,

All the authors that participated in this work would like to take this opportunity to thank you for accepting our manuscript.

Concerning the requested information needed by the journal we have included in the manuscript after the discussion section and prior to the references both the financial and competing interests of all authors along with the contributions of every author.

Furthermore, there is no need figure legend 3 since figure 2 includes 2 images figure 2a and 2b. Figure legend 2 includes comments concerning both topographic images.

Sincerely,

Vasilios F. Diakonis M.D.